Author's response to reviews

Title: Heat shock protein90 in lobular neoplasia of the breast

Authors:

Flora Zagouri (florazagouri@yahoo.co.uk)
Afrodite Nonni (afnonni@med.uoa.gr)
Theodoros N. Sergentanis (tsergentanis@yahoo.gr)
Nikolaos V. Michalopoulos (nmichal@med.uoa.gr)
Andreas C. Lazaris (alazaris@med.uoa.gr)
Efstratios Patsouris (epatsouris@med.uoa.gr)
George C. Zografos (gzografo@med.uoa.gr)

Version: 2 Date: 12 March 2008

Author's response to reviews: see over
To
The Editorial Board of BMC Cancer

Athens, 24/02/2008

Dear Editor,

I am sending for publication to your distinguished Journal the manuscript entitled: “Heat shock protein90 in lobular neoplasia of the breast.”

To date, there is no data regarding the evaluation of Hsp90 in combination with ER-alpha and ER-beta status in patients whose main lesion is lobular neoplasia. To our knowledge, this is the first study to focus on the immunohistochemical expression of Hsps and estrogen receptors alpha and beta in LN cases, establishing their potential prognostic and/or therapeutic implications. Given the originality of the manuscript, we believe that it merits consideration for publication in your Journal.

Yours sincerely,

George C. Zografos,
Assoc. Professor of Surgery
University of Athens